Author's response to reviews

Title: Weight change in control group participants in behavioural weight loss interventions: A systematic review and meta-regression study.

Authors:

Lauren Waters (l.waters1@uq.edu.au)
Alexis St George (alexis.stgeorge@sydney.edu.au)
Tien Chey (tien.chey@sydney.edu.au)
Adrian Bauman (adrian.bauman@sydney.edu.au)

Version: 3 Date: 16 July 2012

Author's response to reviews: see over
July 16, 2012
Editor-in-Chief
BMC Medical Research Methodology

Dear Editor,

Thank you for accepting in principle and giving us the opportunity to revise and resubmit our original review article, entitled "Weight loss in control group participants in behavioural weight loss interventions: A systematic review and meta-regression study" by the authors Lauren Waters (MPH), Alexis St George (PhD), Tien Chey (PhD), and Adrian Bauman (PhD).

The additional comments from reviewer 2 has been addressed, and where indicated, changes have been made to the manuscript. These changes highlighted in the text of the manuscript (Page 8, Paragraph 2, bolded ‘green’)

The manuscript has been revised according to the formatting checklist provided including:

- Reformattting of the references according to journal style
- Addition of tables to the main manuscript
- Figure titles and legends added to the main manuscript in appropriate place
- Addition of Addition File Title to the main manuscript in appropriate place

Sincerely,

Dr Alexis St.George (on behalf of Dr Lauren Waters)
Prevention Research Collaboration
School of Public Health
The University of Sydney
NSW 2006
Australia
Email: alexis.stgeorge@sydney.edu.au
Phone: 9036 3334
Fax: 9036 3184
Phone: +61 7 3365 5163
Fax: +61 7 336 55540